Author's response to reviews

Title: Survival after laparoscopic and open surgery for colon cancer: a comparative, single-institution study

Authors:

Fabio Cianchi (fabio.cianchi@unifi.it)
Giacomo Trallori (gtrallori@interfree.it)
Beatrice Mallardi (b.mallardi@ispo.toscana.it)
Giuseppe Macrì (macrgiuse@virgilio.it)
Maria Rosa Biagini (m.biagini@dfc.unifi.it)
Gabriele Lami (lele08@virgilio.it)
Giampiero Indennitate (endoscopia.ifca@giomi.com)
Siro Bagnoli (bagnolis@aou-careggi.toscana.it)
Andrea Bonanomi (bonanomia@aou-careggi.toscana.it)
Luca Messerini (luca.messerini@unifi.it)
Benedetta Badii (benedettabadii@yahoo.it)
Fabio Staderini (staderini.fabio@gmail.com)
Ileana Skalamera (skalamera.ileana@gmail.com)
Giulia Fiorenza (giulia.fiorenza@yahoo.it)
Giuliano Perigli (giuliano.perigli@unifi.it)

Version: 2 Date: 30 October 2014

Author's response to reviews:

Eden Nonato
Journal Editorial Office
BioMed Central

Dear Dr. Nonato,

Please find a copy of the revised manuscript n. 1780021721147743 entitled, “Survival after laparoscopic and open surgery for colon cancer: a comparative, single-institution study”. It has been amended according to the requested comments. In particular, line numbers and a statement of the approval of the local ethics committee (page 4, lines 11-13) have been added.

We hope the submitted paper will be considered for publication in BMC Surgery.

Sincerely yours,

Fabio Cianchi, M.D.